Author's response to reviews

Title: Do genetic factors protected for early onset lung cancer? - A case-control study before the age of 50 years

Authors:

Albert Rosenberger (arosenb@gwdg.de)
Heike Bickeboller (hbickeb@gwdg.de)
Thomas Illig (illig@gsf.de)
H.-Erich Wichmann (wichmann@gsf.de)

Version: 5 Date: 8 January 2008

Author's response to reviews: see over
Point-by-Point descriptions of changes according to reviewers comments for MS 1397504061543611
“Do genetic factors protected for early onset lung cancer? – A case-control study before the age of 50 years”
Albert Rosenberger et al.

All final comments of the reviewer are listed below and been changes in the tables are done accordingly.

<table>
<thead>
<tr>
<th>Reviewer</th>
<th>Comment</th>
<th>section / page</th>
<th>Comment</th>
<th>Correction</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>1) On Table 3, total study population section, the p-value, OR and 95% CI are still missing for CYP1A1 marker.</td>
<td>Table 3</td>
<td>missing information added: p=0.857 OR=0.68 [0.30-1.51]</td>
<td></td>
</tr>
<tr>
<td></td>
<td>2) Please check the ORs and 95% CI on Table 4 for EPHX1 marker, never and light smokers. They are 1.4 (0.5-4.1) in model I, and 1.1 (0.4-3.0) in model II. I think they should be equal because they refer to the same exposure and reference categories.</td>
<td>Table 4</td>
<td>wrong information replaced: model II: 1.4 (0.5-4.1)</td>
<td></td>
</tr>
</tbody>
</table>